Author’s response to reviews

Title: Retinal vasculitis with Chronic Recurrent Multifocal Osteomyelitis: a case report and review of the literature

Authors:

Victoria Shanmugam (vshanmugam@mfa.gwu.edu)
Marc Phillpotts (mphillpotts@mfa.gwu.edu)
Timothy Brady (tim_brady@gwmail.gwu.edu)
Monica Dalal (mdalal@mfa.gwu.edu)
Shawn Haji-Momenian (shajimomenian@mfa.gwu.edu)
Esma Akin (eakin@mfa.gwu.edu)
Kavita Nataranjan (knataranjan@mfa.gwu.edu)
Sean McNish (mcnish23@gmail.com)
Donald Karcher (dkarcher@mfa.gwu.edu)

Version: 4 Date: 18 Mar 2019

Author’s response to reviews:

Manuscript has been revised as requested. Prior reviewers had asked for the background section to be removed. It has been reinstated.

The local IRB does not consider a case report to be human subjects research, and thus informed consent is not required. This patient did give verbal consent for the report to be published and permitted pathology slides to be included with consent obtained in June 2018.